Author's response to reviews

**Title:** Endometriosis in postmenopausal woman without previous hormonal therapy: a case report

**Authors:**

- manuel garcia-manero (mgmanero@unav.es)
- pedro royo (proyo@alumni.unav.es)
- begoña olartecoecchea (bolarteco@unav.es)
- juan luis alcazar (jlalcazar@unav.es)

**Version:** 4  **Date:** 18 February 2009

**Author's response to reviews:**

She denied current or previous prior history of acyclic pain.

Obstetrical history: two eutocic pregnancy.